Author’s response to reviews

Title: A randomised controlled trial for overweight and obese parents to prevent childhood obesity - Early STOPP (STockholm Obesity Prevention Program)

Authors:

Tanja Sobko (tanja.sobko@ki.se)
Viktoria Svensson (viktoria.svensson@karolinska.se)
Anna Ek (anna.ek@ki.se)
Mirjam Ekstedt (mirjam.ekstedt@esh.se)
Håkan Karlsson (hakan.karlsson@ki.se)
Elin Johansson (Elin.Johansson@ki.se)
Yingting Cao (yingting.cao.937@student.ki.se)
maria Hagströmer (Maria.Hagstromer@ki.se)
Claude Marcus (Claude.Marcus@ki.se)

Version: 4 Date: 5 April 2011

Author’s response to reviews:

Dear Editor,

Thank you for your comments. We have adjusted the manuscript according to your suggestions:

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Tables: we notice that you have included the table as an additional files. If you want this table to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.
- we included the TABLE 1 in the manuscript

References: please cite all references in numerical order in the text. Several citations appear out of sequence, for example: 9-13 and 49.
- We adjusted the references.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details.
- We included the countries in the manuscript.

Design and methods: rename Methods/Design.
- we renamed it in the abstract.

References: please provide the full citation details for reference 4. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references
- we provided full citation details

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
- We ropped the figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
- We checked the manuscript and hope that the manuscript reads better now.

Thank you again,
best regards,
Tanja Sobko